Author’s response to reviews

Title: Prevalence and Causes of Blindness, Visual Impairment Among Different Ethnical Minority Groups In Xinjiang Uygur Autonomous Region, China

Authors:

Yanping Li (5861623@qq.com)
Wenyong Huang (1220792380@qq.com)
Qoyun Qiqige (aoyunqqg@163.com)
Hongwei Zhang (zhwzhw.123@163.com)
Ling Jin (lingjin41@yahoo.com)
Pula Ti (anrabc@126.com)
Jennifer Yip (jennifer.yip@lshtm.ac.uk)
Baixiang Xiao (xiaobaixiang2006@126.com)

Version: 1 Date: 12 Nov 2017

Author’s response to reviews:

Dear Kerry Belbin,

Thanks for your reviewing on my manuscript of BOPH-D-17-00664

Prevalence and Causes of Blindness, Visual Impairment Among Different Ethnical Minority Groups In Xinjiang Uygur Autonomous Region, China.

I here revised the manuscript as reccomended.

1. names and order of Authorships. We do not have changes on that.
2. Initials of the authors' name: changed.
3. Declarations: have been moved to after conclusion as order requirement.
4. Full name of ethical committee is given, also the sample of verbal consent talking.
5. Ethic approval statement is also included in method.
6. List of abbreviation is moved to after conclusion.

7. Background: changed in both abstract and the main document.

Language checked again.